Author’s response to reviews

Title: Title: The learning curve of one anastomosis gastric bypass and its impact as a preceding procedure to Roux-en Y gastric bypass: initial experience of one hundred and five consecutive cases

Authors:

Hung Chieh Lo (carfishcat@yahoo.com.tw)

Version: 2 Date: 12 Feb 2020

Author’s response to reviews:

February 12. 2020

Guangde Tu,
Editor-in-Chief

RE: BSUR-D-19-00971R1
Title: The learning curve of one anastomosis gastric bypass and its impact as a preceding procedure to Roux-en Y gastric bypass: initial experience of one hundred and five consecutive cases

Dear Editor:

Thank you very much for your e-mail dated February 11, 2020, in which you kindly informed us that our manuscript is potentially acceptable for publication in BMC Surgery, and also for your invitation to carried out some essential revisions. I greatly appreciate your kindness. The comments/suggestions were very helpful and we have made revision in the manuscript accordingly.

1. Title Page
We have upload a new main manuscript file that contained title page in page 1 and remove it from the additional files.

2. Consent
Error writing with " Informed written consent was obtained from all individual participants included in the study. "
Had been corrected as “ formal consent was waived due to the retrospective nature of the study”.

3. Conclusion
We have added a “Conclusions” section after the “Discussion” section.

4. Formatting
The suggest to change materials and methods to methods and introduction to background had been made respectively.
5. Clean Manuscript
A clean manuscript with final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours is been upload.

6. English editing service
Certificate for English editing service also attached for your reference.
I sincerely hope that the revised manuscript meets with your approval and the approval of the reviewers, and that it is now acceptable for publication in your esteemed journal.

Sincerely yours,
Hung-Chieh Lo, MD

Division of Trauma and Emergency Surgery, Department of Surgery, Wan Fang Hospital, Taipei Medical University, Taipei, Taiwan
No. 111, Sec. 3, Xinglong Rd., Wenshan Dist., Taipei City 116, Taiwan (R.O.C.)

E-mail: carfishcat@yahoo.com.tw
Tel: +886-2-29307930 ext. 8105   Fax: +886-2-29302448